Author's response to reviews

Title: Psychiatric outpatient consultation for seniors. Perspectives of family physicians, consultants, and patients/family: A descriptive study

Authors:

Mark J Yaffe (mark.yaffe@mcgill.ca)
Francois Primeau (francois.primeau@ssss.gouv.qc.ca)
Jane McCusker (jane.mccusker@mcgill.ca)
Martin G Cole (martin.cole@ssss.gouv.qc.ca)
Eric Belzile (eric.belzile@ssss.gouv.qc.ca)
Nandini Dendukuri (nandini.dendukuri@ssss.gouv.qc.ca)
Michel Elie (michel.elie@ssss.gouv.qc.ca)
Johanne Laplante (johanne.laplante@ssss.gouv.qc.ca)

Version: 2 Date: 4 March 2005

Author's response to reviews:

We thank the reviewer for his comments/suggestions.

(1) Regarding whether satisfaction of patient or family physician was related to expectations for care, we have addressed this in the results section, page 11, 2nd paragraph, last 2 sentences, in which we describe the findings of a new Table, #6, found on page 26. No relationships were found.

(2) Regarding knowledge of the patients who did not show up for their consultation appointment, we now address this in the results section, page 8, paragraph 2, last sentence, where we indicate that we had no information on these patients. We further newly address this issue as a potential study limitation, Page 15, first sentence of paragraph one.

Thank you again for your questions and input. Mark Yaffe